Author’s response to reviews

Title: Improving Emergency Obstetric Referral Systems in Low and Middle Income Countries: A qualitative study in a tertiary health facility in Ghana.

Authors:

Anita Anima Daniels (andaniels@ug.edu.gh; realanidan@yahoo.com)

Aaron Abuosi (aabuosi@ug.edu.gh)

Version: 2 Date: 26 Dec 2019

Author’s response to reviews:

Dear Editor,

Please, the gender of respondents in the quotations have all been removed as suggested.

The manuscript has also been proof read and all track changes have been removed.

We would like to thank you and all the reviewers for the insightful comments and suggestions which have helped shaped this paper. We are also grateful for the opportunity to publish in this journal.

Thank you very much.

Yours sincerely,

Anita Anima Daniels

(Principal Investigator)